Predicted estimates of resting energy expenditure have limited clinical utility in patients with cirrhosis

Ana Teresa Limon-Miro, Clive Douglas Jackson, Tannaz Eslamparast, Hisami Yamanaka-Okumura, Lindsay Dudley Plank, Christiani Jeyakumar Henry, Angela Mary Madden, Livia Garcia Ferreira, Evangelos Kalaitzakis, César Prieto de Frías, Anne Wilkens Knudsen, Leah Gramlich, Maitreyi Raman, Cathy Alberda, Dawn Belland, Vanessa Den Heyer, Puneeta Tandon, Marsha Yvonne Morgan



PII: S0168-8278(22)00018-6

DOI: https://doi.org/10.1016/j.jhep.2022.01.005

Reference: JHEPAT 8576

To appear in: Journal of Hepatology

Received Date: 11 September 2021

Revised Date: 11 January 2022

Accepted Date: 13 January 2022

Please cite this article as: Limon-Miro AT, Jackson CD, Eslamparast T, Yamanaka-Okumura H, Plank LD, Henry CJ, Madden AM, Garcia Ferreira L, Kalaitzakis E, Prieto de Frías C, Knudsen AW, Gramlich L, Raman M, Alberda C, Belland D, Den Heyer V, Tandon P, Morgan MY, Predicted estimates of resting energy expenditure have limited clinical utility in patients with cirrhosis, *Journal of Hepatology* (2022), doi: https://doi.org/10.1016/j.jhep.2022.01.005.

This is a PDF file of an article that has undergone enhancements after acceptance, such as the addition of a cover page and metadata, and formatting for readability, but it is not yet the definitive version of record. This version will undergo additional copyediting, typesetting and review before it is published in its final form, but we are providing this version to give early visibility of the article. Please note that, during the production process, errors may be discovered which could affect the content, and all legal disclaimers that apply to the journal pertain.

© 2022 Published by Elsevier B.V. on behalf of European Association for the Study of the Liver.

Predicted estimates of energy expenditure have limited clinical utility in patients with cirrhosis







Individual **measured** and **predicted** energy expenditure data compared in 900 patients with cirrhosis and 282 healthy controls

Predicted values vs. Measured values Discordance -501 to +548 kcal/kg/24 hr



Prediction Equations (i) Harris-Benedict (ii) Mifflin (iii) Schofield (iv) Henry

Predicted estimates of resting energy expenditure have limited clinical utility in

patients with cirrhosis

Ana Teresa Limon-Miro^{*1}, Clive Douglas Jackson^{*2}, Tannaz Eslamparast¹, Hisami Yamanaka-Okumura³, Lindsay Dudley Plank⁴, Christiani Jeyakumar Henry⁵, Angela Mary Madden⁶, Livia Garcia Ferreira⁷, Evangelos Kalaitzakis^{8,9}, César Prieto de Frías²⁰, Anne Wilkens Knudsen¹¹, Leah Gramlich¹, Maitreyi Raman¹², Cathy Alberda¹³, Dawn Belland¹⁴, Vanessa Den Heyer¹⁴, Puneeta Tandon^{1**}, Marsha Yvonne Morgan^{15**}

- ¹ Department of Medicine, University of Alberta, Edmonton, Canada. <u>analimonmiro@ualberta.ca</u>, <u>eslampar@ualberta.ca</u>, <u>leah.gramlich@ualberta.ca</u>, <u>ptandon@ualberta.ca</u>
- ² Department of Clinical Neurophysiology, Royal Free Hospital, Royal Free London NHS Foundation Trust, London, UK. <u>clive.jackson@nhs.net</u>
- ³ Department of Clinical Nutrition and Food Management, Institute of Biomedical Sciences, Tokushima University Graduate School, Kuramoto-cho, Tokushima, Japan. <u>okumurah@tokushima-u.ac.jp</u>
- ⁴ Department of Surgery, University of Auckland, Auckland, New Zealand, <u>I.plank@auckland.ac.nz</u>
- ⁵ Department of Biochemistry, National University of Singapore, Singapore. jeya henry@sics.a-star.edu.sg
- ⁶ School of Life and Medical Sciences, University of Hertfordshire, Hatfield, UK, <u>a.madden@herts.ac.uk</u>
- ⁷ Graduate Program in Nutrition and Health, Department of Nutrition, Universidade Federal de Lavras, Brazil. <u>livia.ferreira@ufla.br</u>
- ⁸ Gastro Unit, Division of Endoscopy, Copenhagen University Hospital Herlev, Denmark, <u>kalvag@hotmail.com</u>
- ⁹ Department of Gastroenterology, University Hospital of Heraklion, University of Crete, Heraklion, Greece, kalvag@hotmail.com
- ¹⁰ Department of Gastroenterology, Clínica Universidad de Navarra, Pamplona, Spain, <u>cprieto@unav.es</u>
- ¹¹ Gastrounit, Medical Division, Copenhagen University Hospital Hvidovre, Denmark, <u>anne.wilkens.knudsen.01@regionh.dk</u>
- ¹² Department of Medicine, Division of Gastroenterology, University of Calgary, Calgary, Alberta, Canada, <u>mkothand@ucalgary.ca</u>
- ¹³ Royal Alexandra Hospital, Alberta Health Services, Edmonton, Alberta, Canada, <u>cathy.alberda@ahs.ca</u>

1

- ¹⁴ University of Alberta Hospital, Alberta Health Services Nutrition Services, Edmonton, Canada, <u>dawn.belland@albertahealthservices.ca</u>, <u>Vanessa.DenHeyer@albertahealthservices.ca</u>
- ¹⁵ UCL Institute for Liver & Digestive Health, Division of Medicine, Royal Free Campus, University College London, UK, <u>marsha.morgan@ucl.ac.uk</u>

*Denotes co-first authors **Denotes co-senior and co-corresponding authors

Co-corresponding authors:

Marsha Y Morgan, UCL Institute for Liver& Digestive Health, Division of Medicine, Royal Free Campus,

University College London, Hampstead, London, NW3 2PF, UK,

+ 44 (0)20 7433 2866

marsha.morgan@ucl.ac.uk

Puneeta Tandon, Department of Medicine, Division of Gastroenterology (Liver Unit), University of Alberta, Edmonton, Alberta, Canada

+1 (780) 492-7934

ptandon@ualberta.ca

Keywords: energy requirements; ethnicity; indirect calorimetry; liver disease; prediction equations; resting energy expenditure; sex

Electronic Word Count: 6263 Number of Figures & Tables: Figure 1: Tables 6

Conflicts of Interest: None of the authors report a conflict of interest in relation to this work

Financial Support: No relevant financial support

Data Availability Statement: Detailed study data are provided in the supplementary files. The original ethics approvals preclude public sharing of the raw data.

Authors' contributions

Authors' initials	Study concept/ design	Acquisition of data	Analysis & interpretation of data	Drafting of manuscript	Critical revision of manuscript	Statistical analysis	Support	Study supervision
ATL-M			\checkmark	\checkmark	\checkmark			
CDJ	\checkmark		\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
TE		\checkmark	\checkmark	\checkmark	\checkmark			
HY-O		\checkmark			\checkmark			
LDP		\checkmark			✓	Ś.		
CJH		\checkmark			✓			
AMM		\checkmark			 Image: A start of the start of			\checkmark
GF		\checkmark			√			
EK		\checkmark			✓			
CPfF		\checkmark			✓			
AWK		\checkmark		\mathbf{O}	\checkmark			
LG					\checkmark			
MR			0		\checkmark			
CA	\checkmark				\checkmark			
DB					\checkmark			
VDH			2		\checkmark			
MYM	\checkmark	✓	\checkmark	√	\checkmark	\checkmark	✓	\checkmark
РТ	\checkmark	✓ ●	\checkmark	\checkmark	\checkmark		√	\checkmark

Abstract

Background & Aim: Malnutrition is associated with adverse clinical outcomes in patients with cirrhosis. Accurate assessment of energy requirements is needed to optimize dietary intake. Resting energy expenditure (REE), the major component of total energy expenditure, can be measured using indirect calorimetry (mREE) or estimated using prediction equations (pREE). This study assessed the usefulness of predicted estimates of REE in this patient population.

Methods: Individual mREE data were available for 900 patients with cirrhosis (mean [±1SD] age 55.7±11.6 yr; 70% men; 52% south-east Asian) and 282 healthy controls (mean age 36.0±12.8 yr; 52% men; 18% south-east Asian). Metabolic status was classified using thresholds based on the mean±1SD of the mREE in the healthy controls. Comparisons were made between mREE and pREE estimates obtained using the Harris-Benedict, Mifflin, Schofield and Henry equations. Stepwise regression was used to build three new prediction models which included sex, ethnicity, body composition measures, and MELD scores.

Results: The mean mREE was significantly higher in patients than controls when referenced to dry body weight (22.4±3.8 *cf.* 20.8±2.6 kcal/kg/24hr; p<0.001); there were no significant sex differences. The mean mREE was significantly higher in Caucasian than Asian patients (23.1±4.4 *cf.* 21.7±2.9 kcal/kg/24hr; p<0.001). Overall, 37.1% of Caucasians and 25.3% of Asians were classified as hypermetabolic. The differences between mREE and pREE were both statistically and clinically relevant; in the total patient population, pREE estimates ranged from 501 kcal/24hr less to 548 kcal/24hr more than the mREE. Newly-derived prediction equations provided better estimates of mREE but still had limited clinical utility.

Conclusions: Prediction equations do not provide useful estimates of REE in patients with cirrhosis. REE should be directly measured.

Word count: 271

Lay summary

- People with cirrhosis are often malnourished and this has a detrimental effect on outcome.
- Provision of an adequate diet is very important and is best achieved by measuring daily energy requirements and adjusting dietary intake accordingly.
- Prediction equations, which use information on age, sex, weight, and height can be used to
 estimate energy requirements; however, the results they provide are not accurate enough for
 clinical use, particularly as they vary according to sex and ethnicity.

Journal Prevention

Introduction

Malnutrition is common in patients with cirrhosis (1-5) and is a substantial risk factor for bacterial infections, hepatic encephalopathy, hospitalization, and mortality (6-12); improving nutritional status has a beneficial effect on several of these outcome variables (13-19). Several guidelines for the nutritional management of patients with cirrhosis exist and provide recommendations for daily energy intakes based on body weight (20-25). However, the recommended daily energy intakes are inconsistent varying, for example, in patients with compensated disease from 25 kcal/kg (20) to at least 35 kcal/kg (24,25). In addition, several factors mitigate against the use of generic dietary prescriptions *viz:* (i) energy requirement may increase in patients with cirrhosis who develop acute complications (22); (ii) approximately one-quarter to one-third of patients with cirrhosis are hypermetabolic but are not identifiable based on demographic or clinical features (26-30); and, (iii) during the natural course of the disease patients with cirrhosis tend to spontaneously reduce their dietary intake (29) and to reduce their levels of physical activity (31)and this will have compensatory, but not necessarily identifiable effects, on intake requirements.

The inter-patient variability in energy requirement, in patients with cirrhosis, is high and the risk of under- and overfeeding, resulting from a formulaic approach to dietary provision, is likely to be significant. It has, therefore, been recommended that energy requirements should be determined, in this patient population, whenever possible, by indirect calorimetric measurements of resting energy expenditure (mREE) (20-22,24,25). However, this guidance is rarely adopted in clinical practice. Instead estimates of REE from prediction equations (pREE) are used as a substitute even though there are considerable concerns about their accuracy in this patient population (30,32). Further the prediction equations are generic despite substantial evidence of ethnic diversity in energy homeostasis (33-35). Thus, in a recent systematic review of 17 studies involving 1883 patients with cirrhosis, pREE estimates

6

JHEPAI-D-21-01918K3

significantly underestimated mREE in the majority of Caucasian patients while tending to overestimate mREE in patients of south-east Asian origin (32).

The present study utilized individual data from a large cohort of patient with cirrhosis to: a) determine the effects of ethnicity on mREE; b) delineate the prevalence and predictors of hypermetabolism; c) compare REE estimates predicted using the Harris Benedict (36), Mifflin (37), Schofield (38) and Henry (39) equations with mREE; and d) devise new models for predicting both the hypermetabolic state and REE and to determine their clinical utility.

Johnal Prendroo

Subjects and methods

Data sources

The authors of the 17 studies included in a recent systematic review of REE in patients with cirrhosis (32) were approached to determine if they had access to and permission to share individual data on the patients included in their studies and, if available, corresponding data in healthy controls. Additional potential sources of healthy control data were identified from the literature and network contacts.

Individual data extraction

The patient data accessed included: hospitalization status (inpatient vs. outpatient), sex, age, selfreported ethnicity, liver disease aetiology, the presence/absence of fluid retention, height, body weight, estimated dry weight, Pugh's score and Child-Pugh grade (40), Model for End-stage Liver Disease (MELD) score (41), mREE, and, where available, measures of body composition, such as fat free mass (FFM). Patients with ascites, in whom the dry weight had not been estimated, and patients with hepatocellular carcinoma were excluded from the analyses. The healthy controls data accessed included: sex, age, self-reported ethnicity, height, body weight, mREE, and, where available, measures of body composition.

The studies from which data were obtained had been undertaken according to the ethical guidelines of the 1975 Declaration of Helsinki (42) and approved by the appropriate institutional review committees. All participants had provided written informed consent.

Resting energy expenditure

REE was *measured* (mREE) using indirect calorimetry as recommended (32). REE was *predicted* (pREE) using the formulae proposed by Harris-Benedict (36), Mifflin (37), Schofield (38) and Henry (39) (Appendix). mREE and pREE data were expressed both in absolute (kcal/24 hr), and relative (kcal/kg dry body weight/24 hr) terms.

Data curation and statistical analysis

Statistical analyses were performed using R software (43); p-values of <0.05 were considered significant.

Differences in demographic, clinical, anthropometric and metabolic variables between patients with cirrhosis and healthy controls and between patient subgroups were compared using the Wilcoxon rank and χ^2 tests.

Patients' metabolic status was classified using the mean (±1SD) of the mREE in the relevant healthy controls as: hypometabolic (<mean-1SD), normometabolic (mean±1SD), or hypermetabolic (>mean+1SD) (30). The relationships between patients' demographic/clinical features and metabolic status were examined using the Jonckheere-Terpstra test for ordered differences among classes. Logistic regression modelling was used to derive equations suitable for predicting metabolic status, in particular, the hypermetabolic state. Models were built for the total combined control and patient populations and separately, by sex and ethnicity based on both absolute and relative mREE (Supplementary Methods 1).

The interchangeability of mREE and pREE values, identified by the limits of agreement between the two measures and their directionality, were assessed using a modified Bland and Altman approach (44) in which the differences between mREE and pREE were plotted against pREE in order to assess the residuals and hence reduces bias.

The associations between mREE and sex, age, ethnicity, dry weight, height, body mass index (BMI) and MELD score were examined using simple regression. A stepwise regression with forward selection was applied utilizing the available data in the patient and healthy control populations to derive two new prediction models (*Basic and London*) and a third new prediction model (MELD) based on information available in the patient population alone (Supplementary Methods 2). The validity of the newly devised

9

JHEPAI-D-21-01918K3

models was then examined by application to the patient population. The estimates of pREE provided using the new models and the estimates provided by the four standard prediction equations were compared using *standard* Bland and Altman plots (44).

Results

Study subjects

A total of 900 patients with cirrhosis and 282 healthy control subjects were identified and included in the analyses (30,45-51) (Supplementary Table 1).

Patients with cirrhosis

The patients were predominantly middle-aged and male (Table 1). Overall, 61% had viral-related cirrhosis; 69% were either Child-Pugh Grade B or C; 48% were Caucasian while 52% were south-east Asian. The expected male: female differences in anthropometric measurements were observed in both the Caucasian and Asian subpopulations but there was no significant sex-related difference in mean BMI, based on estimated dry body weight (Table 2: Supplementary Table 1). There were, however, a number of significant differences in the demographic and derived variables in the ethnic subpopulations (Table 2). The Asian patients were older (mean [±1SD] 59.8±10.9 *cf*. 51.3±10.8 yr; p<0.001), had a lower mean dry BMI (22.2±3.2 *cf*. 25.5±4.9 kg/m²; p<0.001), were more likely to have HBV/HCV-related cirrhosis (89.7 *cf*. 32.2%; p<0.001), and to have less decompensated disease (MELD score 9.2±4.2 *cf*. 13.7±5.2; p<0.001) (Table 2).

Healthy controls

The control population comprised of 282 healthy people (51.8% men; mean age 36.0 ± 12.8 yr; mean BMI 26.2 ± 5.1 kg/m²) (Table 2). Of these, 50 (17.7%) were Caucasian while the remainder were southeast Asian. The Asian healthy controls were significantly younger than their Caucasian counterparts (mean age 32.8 ± 10.5 *cf.* 50.6 ± 12.4 yr; p<0.001), and significantly shorter (166.3 ± 8.5 *cf.* 169.6 ± 8.4 cm; p<0.05) but were otherwise comparable (Table 2).

Matching of patient and control populations

There were no significant differences in demographic and derived variables between the Caucasian patients and healthy Caucasian controls except for an imbalance in the proportion of men

JHEPAI-D-21-01918K3

(Supplementary Table 2). In contrast, there were several potentially confounding differences between the Asian patients and the healthy Asian controls, notably in sex distribution, BMI and age (Supplementary Table 2). However, the difference in sex distribution was accounted for by correcting the analyses for sex and the differences in weight were largely off-set by correcting REE for body weight. Likewise, the difference in age was non-consequential as there was no significant relationship between age and weight-corrected REE in the healthy Asian controls (correlation coefficient – 0.169; Supplementary Figure 1); in particular there was no significant difference in mREE/kg body weight in healthy Asian controls aged <40 and \geq 40 yr (20.8±2.6 cf. 20.3±2.2 kcal/kg; p = 0.12).

Measured REE

The mean mREE was significantly higher in men than in women in both the total patient and healthy control populations when expressed in absolute terms; the difference was retained in the control population when referenced to body weight (21.3 ± 2.6 *cf.* 20.2 ± 2.5 kcal/kg/24 hr; p<0.001) but was negated in the patient population (22.4 ± 3.7 *cf.* 22.2 ± 3.9 kcal/kg dry body weight/24 hr) (Table 3).

The mean mREE was significantly *lower* in the total patient population compared to the healthy controls when expressed in absolute terms (1447 \pm 354 *cf.* 1499 \pm 293 kcal/24 hr; p<0.02) but was significantly *higher* than in healthy controls when expressed relative to dry body weight (22.4 \pm 3.8 *cf.* 20.8 \pm 2.6 kcal/kg/24 hr (p<0.001) (Table 3).

In the Caucasian population, the mean mREE, expressed in absolute terms, was significantly *higher* in the total patient population than in the healthy controls, and was also significantly higher in the group as a whole, and by sex, relative to dry body weight (Table 3). In the Asian population the mean mREE, expressed in absolute terms, was significantly *lower* in the patients than in the healthy controls both in the group as a whole, and by sex. However, the mean mREE, expressed relative to body weight, was significantly higher in the patients than the healthy controls in the whole group and in women but was comparable in men (Table 3).

JHEPAI-D-21-01918K3

Overall, the mean mREE did not differ significantly between the Caucasian and Asian healthy controls whether expressed in absolute or relative terms. However, the mean mREE in the Caucasian patients was significantly higher than in the Asian patients as a whole and by sex whether expressed in absolute or relative terms (Table 3).

Metabolic status in patients with cirrhosis

The thresholds for determining metabolic status were based on the mean (±1SD) of the mREE in the corresponding healthy control populations (30) (Supplementary Table 3). Significant differences in the classification of metabolic status were observed in the Caucasian and Asian patients, when expressed in absolute terms with approximately 30% of Caucasians classified as *hyper*metabolic but almost 45% of Asians patients classified as *hypo*metabolic (Table 4; Supplementary Tables 4A-F). However, when mREE was expressed relative to dry body weight approximately a quarter to one- third of Caucasian and Asian patients were classified as *hyper*metabolic (Table 4; Supplementary Tables 4G-L). There was no sex- difference in the prevalence of hypermetabolism in the Caucasians (men 33.2%; women 37.1%) but a notable, unexplained, sex-difference in prevalence in the Asians (men 20.5%; women 35.2%; p=0.001). Patients classified as *hyper*metabolic, based on relative REE, were younger, weighed less and had more decompensated liver disease than their *normo*- or *hypo*metabolic counterparts (Supplementary Tables 4G-L).

Predicting metabolic status

Logistic regression modelling was used to determine if it were possible to predict the hypermetabolic state in patients with cirrhosis. The performance of the prediction models based on absolute mREE was fair to moderate (total population: sensitivity 59%, specificity 95%, positive predictive value (PPV) 71%, negative predictive value (NPV) 92%; accuracy 89%) with no notable differences by sex or ethnicity (Supplementary Table 5). The performance of the prediction models based on mREE relative to dry

13

body weight was also fair to moderate (total population: sensitivity 40%, specificity 91%, PPV 64%, NPV 79%; accuracy 77%) again with no notable differences, by sex or ethnicity.

pREE using standard equations

Differences were observed between mREE and pREE estimates, in the healthy controls, relative to the prediction equation, sex and ethnicity (Supplementary Table 6). Overall pREE estimates were within the mean±2SD of mREE in 65.2-72.7% of healthy controls; were under-predicted in 3.9-12.4% and over-predicted in 14.9-29.1%, depending on the equation used. pREE estimates were within the mean±2SD of mREE in 69.9-85.6% of healthy male controls; 79.4-89.0% of healthy female controls; 74.0-78.0% of healthy Caucasian controls and 63.4-69.0% of healthy Asian controls.

Differences were observed between mREE and pREE estimates, in the patients with cirrhosis, again relative to the prediction equation, sex and ethnicity (Table 5). Overall, pREE estimates were within the mean±2SD of mREE in 82.0-86.3% of the patients but were under-predicted in 4.3-13.4% and over-predicted in 3.8-10.9%, depending on the equation used (Table 5). pREE estimates were within the mean±2SD of mREE in 80.1-91.9% of men with cirrhosis and 70.9%-94.0% of women with cirrhosis. In the Caucasian patients as a whole, and by sex, the mean REE estimates provided by all four prediction equations were significantly *lower* than the mean mREE (Table 5). In contrast, in the Asian patients the mean Schofield and Henry pREEs significantly *over-estimated* the mean mREE in the total population and in men (Table 5) while the mean Mifflin and Harris–Benedict pREEs significantly over-estimated the mean mREE in mean and under-estimated mREE in women (data not shown).

Limits of Agreement between the mREE and pREE

Information on data interchangeability was determined by the limits of agreement between measures of mREE and pREE and their directionality.

In the 282 healthy controls, the pREE estimates, provided by the four prediction equations, were from -432 kcal/24 hr less to 301 kcal/24 hr more than the mREE (Supplementary Table 7). In the 900 patients

JHEPAI-D-21-01918K3

with cirrhosis, the pREE estimates, provided by the four equations, were from 501 kcal/24 hr less to 548 kcal/24 hr more than the mREE (Table 6). Similar differences in the limits of agreement between mREE and pREE estimates were observed in the patient populations by sex and by ethnicity (Table 6; Figure 1). The pREE estimates were within the limits of agreement for all four equations in 50% of the patients with cirrhosis. However, these individuals could not be distinguished clinically.

New models to predict REE

Three separate models were built (Supplementary Material 2):

(i) *Basic* model: in the total patient population the limits of agreement, ranged from -427 to 413 kcal/24 hr (Supplementary Table 8); these are marginally narrower than those provided by the four standard prediction equations (Table 6).

(ii) *MELD* model: the limits of agreement, in the total population, ranged from -441 to 441 kcal/24 hr (Supplementary Table 8); these are also marginally narrower than those obtained with the standard prediction equations (Table 6).

(iii) *London* model: the limits of agreement, in the total population, ranged from -398 to 398 kcal/24 hr (Supplementary Table 8); these are the narrowest of the limits of agreement provided by any of the standard and newly devised prediction equations (Table 6).

Discussion

The key findings in the present study were: (i) mREE must be referenced to a measure of body composition, for example, dry body weight; (ii) ethnicity is an important determinant of mREE; (iii) the commonly used prediction equations do not provide REE estimates that can be used interchangeably with mREE and hence have limited clinical utility; and (iv) the major determinants of REE in patients with cirrhosis have yet to be fully delineated limiting the accuracy of any model derived to predict metabolic status and REE.

Expressing mREE in absolute rather than relative terms provides misleading results and inaccurate comparisons. Thus, the mean mREE was significantly *lower* in patients with cirrhosis than in healthy controls, when expressed in absolute terms, but was significantly higher when expressed relative to dry body weight. Likewise, approximately 30% of Caucasian patients were classified as *hyper*metabolic based on absolute mREE values while over 45% of south-east Asian patients were classified as *hypo*metabolic; referencing the mREE to dry body weight identified an excess of *hyper*metabolic patients in both populations.

The unique composition of this cohort of patients allowed the effects of ethnicity on metabolic status in cirrhosis to be explored. The mean mREE was significantly higher in the Caucasian patients than in their south-east Asian counterparts even when allowance was made for differences in BMI. In addition, whilst the frequency of hypermetabolism was comparable in Caucasian men and women (33.6 % *cf.* 37.8%) there was a distinct and unexplained sex-related difference in frequency in the south-east Asians (20.2% *cf.* 36.8%). Similarly, the performance of the prediction equations differed substantially, by ethnicity. Thus, in the Caucasian patients the mean pREE estimates provided by the four prediction equations were significantly lower than the mean mREE. In contrast, in the south-east Asian patients the mean pREEs were either significantly higher than the mean mREE or else did not differ from it significantly. The limits of agreement between the pREE estimates and mREE were considerably wider in the Caucasian patients as were the ranges between the minimum and maximum differences between these variables.

16

JHEPAI-D-21-01918K3

These findings clearly identify ethnicity as a major determinant of mREE and this undoubtedly relates to the well-documented variation in body composition between races, more especially to differences in the size and composition of the FFM (52). Variation in the size of the FFM explains 65 to 90% of the inter-subject variation in mREE in healthy adults (53-55). However, the composition of the FFM is equally as important as REE may be significantly influenced by variation in the ratios of high to low energy-requiring FFM contributors (56-59). In healthy Caucasian adults, skeletal and liver mass are the major FFM contributors to REE (57). Healthy African Americans have significantly lower mREE than their white counterparts largely accounted for by a smaller summed mass of specific high-metabolic-rate organs (60). Asian Indians living in Singapore have a significantly lower mREE than their Chinese counterparts; this difference remains significant after adjustment for total FFM and skeletal muscle mass but is attenuated when adjusted for visceral FFM (61).

Approximately a quarter to one third of patients, in the present study, were classified as hypermetabolic, in line with previous findings (26-30). However, there is little or no consensus on the most accurate method for defining metabolic status in this patient population. Classification has been based on threshold relating to the ratio of mREE to pREE (29), the range of values in healthy controls (28) or, as in this study, the mean + 1SD of healthy control values (30). A universally acceptable, evidence-based classification of metabolic status is clearly required. Patients who are hypermetabolic may be particularly vulnerable to nutritional insult (62,63) and the ability to identify them is a significant unmet need. Several models were built to facilitate prediction of the hypermetabolic state, based on the available individual patient and healthy control data, but their diagnostic performance was only fair to moderate. Future prediction modelling requires accurate information on the factors that underpin the hypermetabolic state (29) which is being actively pursued (50).

The pREE estimates, in the patients with cirrhosis, ranged from 501 kcal/24 hr less to 548 kcal/24 hr more than the mREE rendering them of little value in the clinical setting. This finding is not specific to this patient

JHEPAI-D-21-01918K3

population as the accuracy of the pREE estimates is notably poor in several conditions, for example, obesity, malignancy, and critical illness (64-67). This prediction inaccuracy in disease states has been attributed to the fact that the original equations were developed in cohorts of healthy, non-hospitalized Caucasian individuals of normal weight in first two decades of the 20th century. However, the pREE estimates in the healthy controls, in the present study, were equally as poor ranging from 432 kcal/24 hr less to 301 kcal/kg more than the mREE.

The novel prediction models developed in the present study provided better mREE estimates than those provided by the established prediction equations. Of these the *London* model, which included a measure of FFM derived from skinfold anthropometry (68), provided the best estimates with the narrowest limits of agreement. However, none of the models provided pREE estimates that would be useful clinically.

Currently between 50 to 65% of the variance in mREE in patients with cirrhosis can be explained by demographic and body composition variables (30). Clearly more information is needed on the factors responsible for the remaining variance in REE if clinical useful prediction tools are to be developed (69).

It is likely that referencing mREE not just to FFM but to more specific metabolic components of the FFM would provide more meaningful results and comparisons (69,70). Data on the composition of the FFM in patients with cirrhosis and the contribution of the various FFM components to the variance in REE are lacking and are not easily obtained (57). However, this knowledge might allow development of novel energy expenditure body composition models in the future (69,70).

The present study has a number of strengths: (i) the patient population was large and thus subgroup analyses for differences in potentially important confounders such as sex, body composition, and ethnicity were possible; (ii) the patients were drawn from study centres in Europe, south-east Asia, New Zealand and South America; the findings are, therefore, more generally applicable than if the population were more refined; (iii) a healthy control population was included which allowed for a more precise classification of

JHEPAI-D-21-01918K3

metabolic status than the techniques used in previous publications (48,50,71); and, (iv) the pREE estimate were calculated *de novo* for the four most widely-used equations, for all included patients and healthy controls, hence ensuring consistency.

This study also has a number of limitations: (i) the patients were of either of south-east Asian or Caucasian descent; information was not available on south Asian or African/Afro-Caribbean populations; (ii) the majority of patients had HBV/HCV-related cirrhosis although patients with alcohol-related cirrhosis were well-represented; only 6% had non-alcoholic fatty liver-related cirrhosis, which is not reflective of current clinical practice; (iii) data on the level of functional hepatic reserve, assessed using the Child-Pugh or MELD scores, were available for 51% and 79% of the patient population respectively; however the number of patient available for analysis was still substantial; (iv) data on nutritional status were not generally available and so could not be included - this has particular relevance when determining the appropriateness of recommended energy intakes, by nutritional status (46,72); (v) dry BMI was used as a surrogate for nutritional status but the methods of ascertainment differed between populations; (vi) measures of FFM and skeletal muscle mass were not available for the majority of patients; these would have provided a more robust and less variable tissue reference for REE expression; (vii) the number of Caucasian healthy controls was relatively small but still adequate for analysis, and (vii) it is likely that there were differences in preparation of both the patients and the healthy controls for REE measurements e.g. fasting state; concomitant medication with β -blockers (46) and in measurement techniques e.g. type of indirect calorimeter used, the calibration; the equilibration and measurement periods and the calculation of REE, in particular whether it included an adjustment for 24 hr urinary nitrogen excretion (32).

Conclusions

Prediction equations do not provide estimates of REE that are sufficiently accurate to determine dietary requirements in patients with cirrhosis. These findings further strengthen the recommendation made in recent nutritional guidelines that REE should be directly measured in these patients and not predicted.

19

JHEPAI-D-21-01918K3

Further studies are needed to identify the variables which contribute to the variance in REE in this patient population. It may then be possible to devise accurate prediction models obviating the need for direct REE measurement. If, in addition, evidence-based information on the relationship between mREE and TEE, can be obtained then future recommended weight-based dietary prescriptions could be more securely formulated.

Abbreviations: BMI: Body mass index; FM: fat mass; FFM: fat-free mass; HBV/HCV: hepatitis B/C; ht: height; IPD: individual patient data; MELD: Model for End-stage Liver Disease; mREE: measured resting energy expenditure; NAFLD: non-alcoholic fatty liver disease; NPV: negative predictive value; PPV: positive predictive value; pREE: predicted resting energy expenditure; REE: resting energy expenditure; TEE: total energy expenditure; wt: weight.

Acknowledgements: Dr Ben Vandermeer – Alberta Research Centre for Health Evidence, University of Alberta, Edmonton, AB, Canada for providing guidance on the initial statistical analysis for the manuscript. We are grateful to all the people who took part in these studies.

Appendix

Published equations used to predict REE

Harris-Benedict (36)

Men: 66.47 + (13.75 x wt) + (500.33 x ht) - (6.76 x age)Women: 655.10 + (9.56 x wt) + (184.96 x ht) - (4.68 x age)

Mifflin (37)

Men: (9.99 x wt) + (625 x ht) - (4.92 x age) + 5Women: (9.99 x wt) + (625 x ht) - (4.92 x age) - 161

Schofield (38)

Men: (18-30 yr) = (15.06 x wt) - (10.04 x ht) + 706.12 (>30-60 yr) = (11.48 x wt) - (2.63 x ht) + 877.57 (60+ yr) = (9.09 x wt) + (972.74 x ht) - 834.77

Women: (18-30 yr) = (13.63 x wt) + (283.12 x ht) + 98.28 (>30-60 yr) = (8.13 x wt) + (1.43 x ht) + 844.09 (60+ yr) 5 (7.89 x wt) + (458.39 x ht) + 17.69

Henry (39) Men: (18-30 yr) = (14.4 x wt) + (313 x ht) + 113 (>30-60 yr) = (11.4 x wt) + (541 x ht) - 137 (60+ yr) = (11.4 x wt) + (541 x ht) - 256Women: (18-30 yr) = (10.4 x wt) + (615 x ht) - 282(>30-60 yr) = (8.18 x wt) + (502 x ht) - 11.6

References

- Lautz HU, Selberg O, Körber J, Bürger M, Müller MJ. Protein-calorie malnutrition in liver cirrhosis. Clin Investig 1992;70:478-486.
- Italian Multicentre Cooperative Project on Nutrition in Liver Cirrhosis. Nutritional status in cirrhosis. J Hepatol 1994;21:317-325.
- 3. Loguercio C, Sava E, Sicolo P, Castellano I, Narciso O. Nutritional status and survival of patients with liver cirrhosis: anthropometric evaluation. Minerva Gastroenterol Dietol 1996;42:57-60.
- Maharshi S, Sharma BC, Srivastava S. Malnutrition in cirrhosis increases morbidity and mortality. J Gastroenterol Hepatol 2015;30:1507-1513.
- 5. Park J-H, Kang M, Jun D-W, Kim M, Kwak J-H, Kang B-K. Determining whether low protein intake (<1.0 g/kg) is a risk factor for malnutrition in patients with cirrhosis. J Clin Med 2021;10:2164.
- Shaw BW Jr, Wood RP, Gordon RD, Iwatsuki S, Gillquist WP, Starzl TE. Influence of selected patient variables and operative blood loss on six-month survival following liver transplantation. Semin Liver Dis 1985;5:385-393.
- 7. Pikul J, Sharpe MD, Lowndes R, Ghent CN. Degree of preoperative malnutrition is predictive of postoperative morbidity and mortality in liver transplant recipients. Transplantation 1994;57:469-472.
- Mendenhall C, Roselle GA, Gartside P, Moritz T. Relationship of protein calorie malnutrition to alcoholic liver disease: a reexamination of data from two Veterans Administration Cooperative Studies. Alcohol Clin Exp Res 1995;19:635-641.
- 9. Merli M, Riggio O, Dally L. Does malnutrition affect survival in cirrhosis? PINC (Policentrica Italiana Nutrizione Cirrosi). Hepatology 1996;23:1041-1046.
- 10. Huisman EJ, Trip EJ, Siersema PD, van Hoek B, van Erpecum KJ. Protein energy malnutrition predicts complications in liver cirrhosis. Eur J Gastroenterol Hepatol 2011;23:982-989.
- 11. Kalafateli M, Mantzoukis K, Choi Yau Y, Mohammad AO, Arora S, Rodrigues S, et al. Malnutrition and sarcopenia predict post-liver transplantation outcomes independently of the Model for End-stage Liver Disease score. J Cachexia Sarcopenia Muscle 2017;8:113-121.
- 12. Ney M, Li S, Vandermeer B, Gramlich L, Ismond KP, Raman M, et al. Systematic review with metaanalysis: screening and assessment tools in cirrhosis. Liver Int 2020;40:664-673.

JHEPAI-D-21-01918K3

- Mendenhall C, Bongiovanni G, Goldberg S, Miller B, Moore J, Rouster S, et al. VA Cooperative Study on Alcoholic Hepatitis. III: Changes in protein-calorie malnutrition associated with 30 days of hospitalization with and without enteral nutritional therapy. JPEN J Parenter Enteral Nutr 1985;9:590-596.
- 14. Cabré E, Gonzalez-Huix F, Abad-Lacruz A, Esteve M, Acero D, Fernandez-Bañares F, et al. Effect of total enteral nutrition on the short-term outcome of severely malnourished cirrhotics. A randomized controlled trial. Gastroenterology 1990;98:715-720.
- 15. Kearns PJ, Young H, Garcia G, Blaschke T, O'Hanlon G, Rinki M, et al. Accelerated improvement of alcoholic liver disease with enteral nutrition. Gastroenterology 1992;102:200-205.
- 16. Hirsch S, Bunout D, de la Maza P, Iturriaga H, Petermann M, Icazar G, et al. Controlled trial on nutrition supplementation in outpatients with symptomatic alcoholic cirrhosis. JPEN J Parenter Enteral Nutr 1993;17:119-124.
- 17. Mendenhall CL, Moritz TE, Roselle GA, Morgan TR, Nemchausky BA, Tamburro CH, et al. A study of oral nutritional support with oxandrolone in malnourished patients with alcoholic hepatitis: results of a Department of Veterans Affairs cooperative study. Hepatology 1993;17:564-576.
- Campillo B, Bories PN, Leluan M, Pornin B, Devanlay M, Fouet P. Short-term changes in energy metabolism after 1 month of a regular oral diet in severely malnourished cirrhotic patients. Metabolism 1995;44:765-770.
- 19. Sriram K, Sulo S, VanDerBosch G, Partridge J, Feldstein J, Hegazi RA, et al. A comprehensive nutritionfocused quality improvement programme reduces 30-day readmissions and length of stay in hospitalized patients. JPEN J Parenter Enteral Nutr 2017;41:384-391.
- 20. Plauth M, Merli M, Kondrup J, Weimann A, Ferenci P, Müller MJ, et al. ESPEN guidelines for nutrition in liver disease and transplantation. Clin Nutr 1997;16:43-55.
- 21. Plauth M, Cabré E, Riggio O, Assis-Camilo M, Pirlich M, Kondrup J, et al. ESPEN Guidelines on enteral nutrition: liver disease. Clin Nutr 2006;25:285-294.
- 22. Plauth M, Bernal W, Dasarathy S, Merli M, Plank LD, Schutz T, et al. ESPEN guideline on clinical nutrition in liver disease. Clin Nutr 2019;38:485-521.

- 23. Amodio P, Bemeur C, Butterworth R, Cordoba J, Kato A, Montagnese S, et al. The nutritional management of hepatic encephalopathy in patients with cirrhosis: International Society for Hepatic Encephalopathy and Nitrogen Metabolism Consensus. Hepatology 2013;58:325-336.
- 24. European Association for the Study of the Liver. EASL Clinical Practice Guidelines on nutrition in chronic liver disease. J Hepatol 2019;70:172-193.
- 25. Lai JC, Tandon P, Bernal W, Tapper EB, Ekong U, Dasarathy S, et al. Malnutrition, frailty, and sarcopenia in patients with cirrhosis:2021 Practice Guidance by the American Association for the Study of Liver Diseases. Hepatology 2021;74:1611-1644.
- 26. Shanbhogue RLK, Bistrian BR, Jenkins RL, Jones C, Benotti P, Blackbum GL. Resting energy expenditure in patients with end stage liver disease and normal population. JPEN 1987; 11: 305-308
- 27. Schneeweiss BN, Graninger W, Ferenci P, Eichinger S, Grimm G, Schneider B, et al. Energy metabolism in patients with acute and chronic liver disease. Hepatology 1990;11:387-393.
- 28.Müller M J, Lautz H U, Plogmann B, Bürger M, <u>Körber</u> J, Schmidt FW. Energy expenditure and substrate oxidation in patients with cirrhosis: The impact of cause, clinical staging and nutritional state. Hepatology 1992,15:782-794.
- 29. Müller MJ, Böttcher J, Selberg O, Weselmann S, Böker KH, Schwarze M, et al. Hypermetabolism in clinically stable patients with liver cirrhosis. Am J Clin Nutr 1999;69:1194-1201.
- 30. Madden AM, Morgan MY. Resting energy expenditure should be measured in patients with cirrhosis, not predicted. Hepatology 1999;30:655-664.
- 31. Müller MJ, Dettmer A, Tettenborn M, Radoch E, Fichter J, Wagner TO, et al. Metabolic, endocrine, haemodynamic and pulmonary responses to different types of exercise in individuals with normal or reduced liver function. Eur J Appl Physiol Occup Physiol 1996;74:246-257.
- 32. Eslamparast T, Vandermeer B, Raman M, Gramlich L, Den Heyer V, Belland D, et al. Are predictive energy expenditure equations accurate in cirrhosis? Nutrients 2019;11:334
- Donahoo WT, Levine JA, Melanson EL. Variability in energy expenditure and its components. Curr Opin Clin Nutr Metab Care 2004;7:599-605.

- 34. Douglas CC, Lawrence JC, Bush NC, Oster RA, Gower BA, Darnell BE. Ability of the Harris Benedict formula to predict energy requirements differs with weight history and ethnicity. Nutr Res 2007;27:194-199.
- 35. Hasson RE, Howe CA, Jones BL, Freedson PS. Accuracy of four resting metabolic rate prediction equations: effects of sex, body mass index, age, and race/ethnicity. J Sci Med Sport 2011;14:344-351.
- Harris JA, Benedict FG. A Biometric Study of Basal Metabolism in Man. Washington, DC Carneige Institute of Washingtom, 1919: Publication no. 279.
- Mifflin MD, St Jeor ST, Hill LA, Scott BJ, Daugherty SA, Koh YO. A new predictive equation for resting energy expenditure in healthy individuals. Am J Clin Nutr 1990;51:241-247.
- Schofield WN. Predicting basal metabolic rate, new standards and review of previous work. Hum Nutr Clin Nutr 1985;39 (Suppl 1):5-41.
- Henry CJK. Basal metabolic rate studies in humans: measurement and development of new equations. Public Health Nutr 2005;8:1133-1152.
- 40. Pugh RN, Murray-Lyon IM, Dawson JL, Pietroni MC, Williams R. Transection of the oesophagus for bleeding oesophageal varices. Br J Surg 1973;60:646-649.
- 41. Malinchoc M, Kamath PS, Gordon FD, Peine CJ, Rank J, ter Borg PC. A model to predict poor survival in patients undergoing transjugular intrahepatic portosystemic shunts. Hepatology 2000;31:864-871.
- 42. World Medical Association Declaration of Helsinki: ethical principles for medical research involving human subjects. JAMA 2013;310:2191-2194.
- 43. Team RC. R: A language and environment for statistical computing Vienna, Austria: R Foundation for Statistical Computing; 2020 [Available from: <u>https://www.R-project.org/</u>]
- 44. Bland JM, Altman DG. Statistical methods for assessing agreement between two methods of clinical measurement. Lancet 1986;1:307-310.
- 45. Kalaitzakis E, Bosaeus I, Ohman L, Björnsson E. Altered postprandial glucose, insulin, leptin, and ghrelin in liver cirrhosis: correlations with energy intake and resting energy expenditure. Am J Clin Nutr 2007;85:808-815.

- 46. Peng S, Plank LD, McCall JL, Gillanders LK, McIlroy K, Gane EJ. Body composition, muscle function, and energy expenditure in patients with liver cirrhosis: a comprehensive study. Am J Clin Nutr 2007;85:1257-1266.
- 47. Ferreira LG, Santos LF, Silva TR, Anastácio LR, Lima AS, Correia MI. Hyper- and hypometabolism are not related to nutritional status of patients on the waiting list for liver transplantation. Clin Nutr 2014;33:754-760.
- 48. Teramoto A, Yamanaka-Okumura H, Urano E, Nakamura-Kutsuzawa T, Sugihara K, Katayama T, et al. Comparison of measured and predicted energy expenditure in patients with liver cirrhosis. Asia Pac J Clin Nutr 2014;23:197-204.
- 49. Knudsen AW, Krag A, Nordgaard-Lassen I, Frandsen E, Tofteng F, Mortensen C, et al. Effect of paracentesis on metabolic activity in patients with advanced cirrhosis and ascites. Scand J Gastroenterol 2016;51:601-609.
- 50. Prieto-Frías C, Conchillo M, Payeras M, Inarrairaegui M, Davola D, Fruhbeck G, et al. Factors related to increased resting energy expenditure in men with liver cirrhosis. Eur J Gastroenterol Hepatol 2016;28:139-145.
- 51. Camps SG, Wang NX, Tan WS, Henry CJ. Estimation of basal metabolic rate in Chinese: are the current prediction equations applicable? Nutr J 2016;15:79.
- 52. Wulan SN, Westerterp KR, Plasqui G. Ethnic differences in body composition and the associated metabolic profile: a comparative study between Asians and Caucasians. Maturitas 2010;65:315-319.
- 53. Garby L, Garrow JS, Jørgensen B, Lammert O, Madsen K, Sørensen P, et al. Relation between energy expenditure and body composition in man: specific energy expenditure in vivo of fat and fat-free tissue. Eur J Clin Nutr 1988;42:301-305.
- 54. Astrup A, Buemann B, Christensen NJ, Madsen J, Gluud C, Bennett P, et al. The contribution of body composition, substrates, and hormones to the variability in energy expenditure and substrate utilization in premenopausal women. J Clin Endocrinol Metab 1992;74:279-286.
- 55. Tataranni PA, Ravussin E. Variability in metabolic rate: biological sites of regulation. Int J Obes Relat Metab Disord 1995;19(Suppl 4):S102-S106.
- Nelson KM, Weinsier RL, Long CL, Schutz Y. Prediction of resting energy expenditure from fat-free mass and fat mass. Am J Clin Nutr 1992;56:848-856.

- 57. Illner K, Brinkmann G, Heller M, Bosy-Westphal A, Müller MJ. Metabolically active components of fat free mass and resting energy expenditure in nonobese adults. Am J Physiol Endocrinol Metab 2000;278:E308-E315.
- Ravussin E, Bogardus C. Relationship of genetics, age, and physical fitness to daily energy expenditure and fuel utilization. Am J Clin Nutr 1989;49(5 Suppl):968-975.
- 59. Weinsier RL, Schutz Y, Bracco D. Reexamination of the relationship of resting metabolic rate to fat-free mass and to the metabolically active components of fat-free mass in humans. Am J Clin Nutr 1992;55:790-794.
- 60. Gallagher D, Albu J, He Q, Heshka S, Boxt L, Krasnow N, et al. Small organs with a high metabolic rate explain lower resting energy expenditure in African American than in white adults. Am J Clin Nutr 2006;83:1062-1067.
- 61. Song LL, Venkataraman K, Gluckman P, Chong YS, Chee MW, Khoo CM, et al. Smaller size of high metabolic rate organs explains lower resting energy expenditure in Asian-Indian than Chinese men. Int J Obes 2016;40:633-638.
- 62. Verboeket-van de Venne WP, Westerterp KR, van Hoek B, Swart GR. Energy expenditure and substrate metabolism in patients with cirrhosis of the liver: effects of the pattern of food intake. Gut 1995;36:110-116.
- 63. Müller MJ. Hepatic energy and substrate metabolism: a possible metabolic basis for early nutritional support in cirrhotic patients. Nutrition 1998;14:30-38.
- 64. Madden AM, Mulrooney HM, Shah S. Estimation of energy expenditure using prediction equations in overweight and obese adults: a systematic review. J Hum Nutr Diet 2016;29:458-476.
- 65. Reeves MM, Battistutta D, Capra S, Bauer J, Davies PS. Resting energy expenditure in patients with solid tumors undergoing anticancer therapy. Nutrition 2006;22:609-615.
- 66. De Waele E, Opsomer T, Honoré PM, Diltoer M, Mattens S, Huyghens L, et al. Measured versus calculated resting energy expenditure in critically ill adult patients. Do mathematics match the gold standard? Minerva Anestesiol 2015;81:272-282.
- 67. Picolo MF, Lago AF, Menegueti MG, Nicolini EA, Basile-Filho A, Nunes AA, et al. Harris-Benedict equation and resting energy expenditure estimates in critically ill ventilator patients. Am J Crit Care 2016;25:e21-9.

- 68. Durnin JV, Womersley J. Body fat assessed from total body density and its estimation from skinfold thickness: measurements on 481 men and women aged from 16 to 72 years. Br J Nutr 1974;32:77-97.
- 69. Heymsfield SB, Thomas D, Bosy-Westphal A,, Shen W, Peterson CM, Müller MJ. Evolving concepts on adjusting human resting energy expenditure measurements for body size. Obes Rev 2012;13:1001–1014
- 70. Heymsfield SB, Peterson CM, Bourgeois B, Thomas DM, Gallagher D, Strauss B, et al. Human energy expenditure: advances in organ-tissue prediction models. Obes Rev 2018;19:1177-1188.
- 71. Tajika M, Kato M, Mohri H, Miwa Y, Kato T, Ohnishi H, et al. Prognostic value of energy metabolism in patients with viral liver cirrhosis. Nutrition 2002;18:229-234.
- 72. Kalman DR, Saltzman JR. Nutrition status predicts survival in cirrhosis. Nutr Rev 1996;54:217-219.

Journal Press

Legends to Figure



Figure 1: The degree of agreement between measured REE in 900 patients with cirrhosis and the estimates provided by the Harris–Benedict (36), Mifflin (37), Schofield (38) and Henry (39) equations. The blue horizontal line is the line of identity; the horizontal green lines represent the mean and limits of agreement of the data; the difference between the line of identity and the mean represents the bias; the red line represents the best fit through the data set

Variable	Number	Population			
		Total	Men	Women	
Sex (n: %)	900		632 (70.2)	268 (29.8)	
$\Lambda g_{0}(yr)$	000	55.7 ± 11.6	56.4 ± 11.3	54.3 ± 12.3	
Age (yr)	900	(20.0 - 84.0)	(20.0 - 84.0)	(20.0 - 78.0)	
lleicht (om)	900	165.5 ± 9.3	168.8 ± 7.7	$157.5 \pm 8.0^{***}$	
Height (CIII)		(134.0 - 190.0)	(146.0 - 190.0)	(134.0 - 178.0)	
Scale weight (kg)	000	67.0 ± 15.6	69.7 ± 15.7	60.8 ± 13.7***	
Scale weight (kg)	900	(33.5 – 131.8)	(40.2 - 131.8)	(33.5 - 105.5)	
Ascites (n: %)	900	193 (21.4)	127 (20.1)	66 (24.6)***	
	000	65.3 ± 14.9	68.0 ± 14.8	$58.9 \pm 13.1^{***}$	
Dry weight (kg)	900	(33.5 – 131.8)	(40.2 - 131.8)	(33.5 - 105.5)	
	000	23.7 ± 4.5	23.8 ± 4.5	23.7 ± 4.5	
Dry Bivii (kg/m²)	900	(13.1 - 41.1)	(14.8 - 41.1)	(13.1 - 40.7)	
Cirrhosis aetiology (n: %)		X	610 (71.1)	248 (28.9)	
Alcohol		198 (23.1)	142 (23.3)	56 (22.6)	
NAFLD	050	48 (5.6)	34 (5.6)	14 (5.6)	
HBV	858	167 (19.5)	136 (22.3)	31 (12.5)	
HCV		356 (41.5)	258 (42.3)	98 (39.5)	
Other		89 (10.4)	40 (6.6)	49 (19.8)	
			497 (70.3%)	210 (29.7%)	
MELD score	707	12.0 ± 5.3	11.6 ± 5.0	$12.8 \pm 5.8^{*}$	
		(6.0 - 36.0)	(6.0 - 33.0)	(6.0 - 36.0)	
			251 (66.2%)	128 (33.8%)	
Pugh's score	379	8 ± 2	8.0 ± 2.4	8.2 ± 2.3	
		(5 - 14)	(5.0 - 14.0)	(5.0 - 13.0)	
Child-Pugh Grade (n: %)			310 (67.4)	150 (32.6)	
Α		143 (31.1)	104 (33.5)	39 (26.0)	
B	460	186 (40.4)	118 (38.1)	68 (45.3)	
c		131 (28.5)	88 (28.4)	43 (28.7)	
Hospital inpatient (n: %)	900		397 (62.8)	145 (54.1)	

Table 1: Measured and derived demographic variables and hospital status in the total patient population

BMI: body mass index; NAFLD: non-alcoholic fatty liver disease; HBV/HCV: hepatitis B/C; MELD: model of end stage liver disease

Data are presented as mean ± 1SD (range) or as number (%)

Significance of the difference between men and women: * p < 0.05; ** p < 0.01; *** p<0.001

Variable			Patients	with Cirrhosis		
			(n = 900)	. .	
		Caucasian			Asian	
	All (n = 429)	IVIEN	(n - 1/3)	All (n = 471)	Men (n = 346	(n = 125)
Sex (% men)	(11 - 423)	66.7%	(11 - 143)	(11 - 47 1)	73.5%*	(11 - 125)
Age (yr)	51.3 ± 10.8	51.3 ± 10.0	51.3 ± 12.2	59.8 ± 10.9***	60.5 ± 10.6***	57.7 ± 11.5 ^{***}
Height (cm)	169.0 ± 9.1	173.0 ± 7.0	161.0 ± 7.3	$162.3 \pm 8.3^{***}$	$165.4 \pm 6.4^{***}$	$153.6 \pm 6.8^{***}$
Scale weight (kg)	76.2 ± 15.1	80.6 ± 14.3	67.4 ± 12.6	58.7 ± 10.7***	60.7 ± 10.0***	53.2 ± 10.6***
Ascites (%)	183 (42.7%)	120 (42.0%)	63 (44.1%)	10 (2.1%)***	7 (2.0%)***	3 (2.4%)***
Dry weight (kg)	72.8 ± 15.4	77.2 ± 14.7	64.0 ± 13.0	58.5 ± 10.5***	$60.5 \pm 9.8^{***}$	$53.0 \pm 10.4^{***}$
Dry BMI (kg/m ²)	25.5 ± 4.9	25.9 ± 4.9	24.7 ± 4.9	$22.2 \pm 3.2^{***}$	$22.1 \pm 3.1^{***}$	$22.4 \pm 3.6^{***}$
Aetiology (% viral)	32.2%	36.4%	23.8%	89.7%***	89.5%***	90.5%***
MELD score	13.7 ± 5.2	13.6 ± 5.0	14.1 ± 5.5	$9.2 \pm 4.2^{***}$	9.0 ± 3.7***	$10.1 \pm 5.5^{***}$
Variable		JI.	Healt	hy Controls n = 282)		-
		Caucasian			Asian	
	All	Men	Women	All	Men	Women
	(n = 50)	(n = 25)	(n = 25)	(n = 232)	(n = 121)	(n = 111)
Sex (% men)		50.0%			52.2%	
Age (yr)	50.6 ± 12.4	51.4 ± 12.8	49.9 ± 12.2	$32.8 \pm 10.5^{***}$	$32.3 \pm 9.9^{***}$	$33.4 \pm 11.2^{***}$
Height (cm)	169.6 ± 8.4	175.4 ± 6.9	163.8 ± 5.1	$166.3 \pm 8.5^*$	$172.0 \pm 6.0^{*}$	$160.1 \pm 6.2^{**}$
Scale weight (kg)	74.0 ± 12.9	79.6 ± 11.0	68.4 ± 12.3	72.9 ± 16.8	79.2 ± 14.9	66.1 ± 16.1
BMI (kg/m²)	25.7 ± 3.9	25.9 ± 3.2	25.5 ± 4.5	26.3 ± 5.4	26.8 ± 4.9	25.7 ± 5.8

Table 2: Comparison of measured and derived variables in the patients with cirrhosis and healthy controls, by sex and ethnicity

BMI; body mass index

Data are presented as mean ± 1SD (range) or as number (%)

Significance of the difference between Caucasian and Asian populations: *p <0.05; **p<0.01; *** p<0.001

Table 3: Measured REE in patients and healthy controls expressed in absolute and relative terms, by sex and ethnicity

Veriable		Patients		Healthy controls			
Variable	Total	Caucasian	Asian	Total	Caucasian	Asian	
			Total				
	(n = 900)	(n = 429)	(n = 471)	(n = 282)	(n = 50)	(n = 232)	
REE: kcal/24hr	1447 ± 354^^	1652 ± 349^	1261 ± 236^^^,+++	1499 ± 293	1542 ± 283	1489 ± 295	
REE: kcal/kg/24hr	22.4 ± 3.8^^^	23.1 ± 4.4^^	21.7 ± 2.9^^^,+++	20.8 ± 2.6	21.0 ± 3.0	20.7 ± 2.5	
	-		Men		-		
	(n = 632)	(n = 286)	(n = 346)	(n = 146)	(n = 25)	(n = 121)	
REE: kcal/24hr	1515 ± 362^^	1765 ± 333	1309 ± 230^^^,+++	1668 ± 248	1691 ± 254	1663 ± 248	
REE: kcal/kg/24hr	22.4 ± 3.7^^^	23.2 ± 4.4^	21.8 ± 2.9***	21.3 ± 2.6	21.4 ± 3.0	21.3 ± 2.5	
		2	Women				
	(n = 268)	(n = 143)	(n = 125)	(n = 136)	(n = 25)	(n = 111)	
REE: kcal/24hr	1287 ± 276***	1426 ± 260***	1128 ± 198^^^***,+++	1317 ± 221***	1393 ± 229***	1300 ± 217***	
REE: kcal/kg/24hr	22.2 ± 3.9^^^	22.7 ± 4.5^	21.5 ± 3.1^^^,++	20.2 ± 2.5***	20.6 ± 2.9	20.1 ± 2.4***	

REE: resting energy expenditure

Data are presented as mean \pm 1SD in kcal/24 hr or kcal/kg dry body weight/24 hr

Significance of difference between patients and healthy controls: ^ p < 0.05, ^^ p < 0.02, ^^^ p < 0.001

Significance of the difference between men and women: ***p <0.001

Significances of the difference between Caucasians and Asians populations ⁺⁺ p< 0.02; ⁺⁺⁺p < 0.001

Table 4: Metabolic status in the Caucasian and Asian patients with cirrhosis, by sex expressed

Population	Number	Metabolic status*						
Population	Number	Hypometabolic	Normometabolic	Hypermetabolic				
	REE kcal/24 hr							
	Total (n:%)							
Total	900	251 (27.9)	491 (54.6)	158 (17.6)				
Men	632	290 (45.9)	250 (39.6)	92 (14.6)				
Women	268	74 (27.6)	145 (54.1)	49 (18.3)				
		Caucasian (n	:%)					
Total	429	49 (11.4)	254 (59.2)	126 (29.4)				
Men	286	50 (17.5)	157 (54.9)	79 (27.6)				
Women	143	16 (11.2)	94 (65.7)	33 (23.1)				
	Asian (n:%)							
Total	471	210 (44.6)	246 (52.2)	15.(3.2)				
Men	346	245 (70.8)	98 (28.3)	3 (0.9)				
Women	125	59 (47.2)	59 (47.2)	7 (5.6)				
	REE kcal/kg dry body weight/24 hr							
		Total (n:%	6)					
Total	900	95 (10.6)	519 (57.7)	286 (31.8)				
Men	632	86 (13.6)	369 (58.4)	177 (28.0)				
Women	268	30 (11.2)	135 (50.4)	103 (38.4)				
		Caucasian (n	:%)					
Total	429	45 (10.5)	225 (52.4)	159 (37.1)				
Men	286	32 (11.2)	159 (55.6)	95 (33.2)				
Women	143	17 (11.9)	73 (51.0)	53 (37.1)				
	Asian (n:%)							
Total	471	50 (10.6)	302 (64.1)	119 (25.3)				
Men	346	51 (14.7)	224 (64.7)	71 (20.5)				
Women	125	13 (10.4)	68 (54.4)	44 (35.2)				

in absolute and relative terms

*Thresholds for determination of metabolic status were derived from the mean ± 1SD of the relevant healthy control population

Table 5: Distribution of predicted REE relative to measured REE in 900 patients with cirrhosis,

Method	REE Measured vs. Predicted			ed
	(kcal/24 hr)	Under	Within	Over
Measured	1447 ± 354			
Harris-Benedict	1403 ± 258++	95 (10.6%)	738 (82.0%)	67 (7.4%)
Mifflin	1368 ± 249+++	121 (13.4%)	745 (82.8%)	34 (3.8%)
Schofield	1469 ± 256	39 (4.3%)	763 (84.8%)	98 (10.9%)
Henry	1420 ± 243	243 61 (6.8%) 777 (86.39		62 (6.9%)
	Ν	Леп (n = 632)		-
Measured	1515 ± 362		0	
Harris-Benedict	1466 ± 267++	84 (13.3%)	506 (80.1%)	42 (6.6%)
Mifflin	1463 ± 208++	36 (5.7%)	581 (91.9%)	15 (2.4%)
Schofield	1550 ± 251+	28 (4.4%)	550 (87.0%)	54 (8.5%)
Henry	1503 ±225	18 (2.8%)	580 (91.8%)	34 (5.4%)
	W	omen (n = 268)		
Measured	1287 ± 276			
Harris-Benedict	1256 ± 157	6 (2.2%)	249 (92.9%)	13 (4.9%)
Mifflin	1145 ± 188***	73 (27.2%)	190 (70.9%)	5 (1.9%)
Schofield	1279 ± 143	6 (2.2%)	252 (94.0%)	10 (3.7%)
Henry	1225 ± 154++	15 (5.6%)	247 (92.2%)	6 (2.2%)
	Са	ucasian (n = 429)		
Measured	1652 ± 349			
Harris-Benedict	1540 ± 252+++	80 (18.6%)	333 (77.6%)	16 (3.7%)
Mifflin	1481 ± 244 ⁺⁺⁺	107 (24.9%)	317 (73.9%)	5 (1.2%)
Schofield	1591 ± 250++	62 (14.5%)	344 (80.2%)	23 (5.4%)
Henry	1535 ± 244+++	84 (19.6%)	332 (77.4%)	13 (3.0%)
	ŀ	Asian (n = 471)		
Measured	1261 ± 236			
Harris-Benedict	1279 ± 193	24 (5.1%)	379 (80.5%)	68 (14.4%)
Mifflin	1265 ± 205	66 (14.0%)	345 (73.2%)	60 (12.7%)
Schofield	1359 ± 207+++	7 (1.5%)	323 (68.6%)	141 (29.9%)
Henry	1316 ± 188+++	20 (4.2%)	365(77.5%)	86 (18.3%)

by sex and ethnicity

Data are presented as mean ± 1SD or as number (%)

Significance of the difference between mREE and pREE estimates: $^{++} p < 0.01$; $^{+++} p < 0.001$

Method	REE	Mean/SD	Limits of	Range [^]	
	(kcal/24 hr)	of Differences	Agreement		
		Total (n = 900)			
Measured	1447 ± 354				
Harris-Benedict	1403 ± 258++	44 + 224	-404 to 492	-565 to 1361	
Mifflin	1368 ± 249***	79 + 234	-390 to 548	-520 to 1460	
Schofield	1469 ± 256	-22 + 239	-501 to 456	-688 to 1319	
Henry	1420 ±243	27 + 234	-441 to 495	-602 to 1390	
		Men (n = 632)	X		
Measured	1515 ± 362				
Harris-Benedict	1466 ± 267++	49 + 234	-419 to 518	-565 to 1361	
Mifflin	1463 ± 208++	52 + 246	-440 to 545	-520 to 1460	
Schofield	$1550 \pm 251^+$	-35 + 250	-535 to 465	-688 to 1319	
Henry	1503 ±225	12+ 245	-477 to 501	-602 to 1390	
		Women (n = 268)			
Measured	1287 ± 276				
Harris-Benedict	1256 ± 157	31 + 197	-362 to 425	-542 to 706	
Mifflin	1145 ± 188+++	142 + 190	-238 to 522	-421 to 835	
Schofield	1279 ± 143	7 + 208	-409 to 424	-648 to 665	
Henry	1225 ± 154++	62 + 203	-343 to 467	-581 to 757	
		Caucasian (n = 429)			
Measured	1652 ± 349				
Harris-Benedict	1540 ± 252+++	112 + 263	-414 to 638	-565 to 1361	
Mifflin	1481 ± 244+++	171 + 263	-356 to 698	-515 to 1460	
Schofield	1591 ± 250++	61 + 268	-475 to 597	-688 to 1319	
Henry	1535 ± 244+++	117 + 266	-414 to 648	-602 to 1390	
		Asian (n = 471)			
Measured	1261 ± 236				
Harris-Benedict	1279 ± 193	-18 + 157	-333 to 297	-524 to 510	
Mifflin	1265 ± 205	-5 + 165	-335 to 325	-520 to 499	
Schofield	1359 ± 207***	-69 + 178	-455 to 258	-601 to 528	
Henry	1315 ± 188***	-55 + 162	-379 to 269	-538 to 487	

Table 6: Degree of agreement between measured REE and the estimates provided by the four prediction equations in patients with cirrhosis, by sex and ethnicity

Data are presented as mean ± 1SD

^Range: minimum/maximum difference between measured and predicted REE

Significance of the difference between measured REE and predicted estimates; ⁺⁺ p < 0.01; ⁺⁺⁺ p < 0.001

Manuscript JHEPAT-D-21-01918

Highlights

- Malnutrition has a significant negative effect on outcome in patients with cirrhosis
- Accurate information on daily energy requirements is essential for effective nutritional management
- Prediction equations provide estimates of daily energy expenditure but these vary significantly by ethnicity and can range from 501 kcal less to 548 kcal more than measured values
- Where available, resting energy expenditure should be measured using indirect calorimetry

Jonuly